Author’s response to reviews

Title: The impact of glaucoma on quality of life in Ethiopia: a case-control study

Authors:

Fisseha Ayele (drfisseha@yahoo.com;drfissehaa@gmail.com)
Banchamelak Zeraye (bzeraye@yahoo.com)
Yared Wolde (assefabi2003@yahoo.com)
kbrom Legesse (kbromu@gmail.com)
Telake Azale (telakea@yahoo.com)
Matthew Burton (Matthew.Burton@lshtm.ac.uk)

Version: 3 Date: 21 Oct 2017

Author’s response to reviews:

Response to the comments

Dear editors, thanks for the comments. Here are our responses

1. Please rename your Introduction heading so that it appears as Background.
   - “Introduction” on page 2 line 1 is now renamed as “Background”

2. As indicated by our submission guidelines, please include a Conclusions heading in the text of your manuscript, as both a Conclusion and a Discussion section should be provided.
   - “Conclusion” section is now included on page 12 after the end of discussion

3. In your Ethics approval section, please clarify whether informed consent obtained from participants was written or verbal.
   - It is already mentioned on the “Ethical consideration” section of the manuscript (Page 5, paragraph 1, lines 2 and 3) that patients are enrolled in the study if they gave verbal consent. But we have now included a statement “Individuals were enrolled into the study if they gave informed verbal consent.” on the “Declaration” section on page 13 line 8
4. Thank you for including a Funding heading for the Declarations section. Please also state the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

- We have included a statement “The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.” On page 13 line 14

5. Thank you for providing an Availability of data and materials subheading in your Declarations section. Please describe where the raw data generated/analysed during your study can be found. If you wish to share the data through correspondence then please include the statement “The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.” If you do not wish to share your data, please state this along with the reason.

- A statement “The datasets analysed during the current study are available from the corresponding author on reasonable request.” is included on page 13 line 11

6. Please reformat your in-text citations according to our submission guidelines:

- All the reference numbers in the text are now in brackets

7. Please remove the response to reviewers from the file inventory, as it is no longer needed at this stage of the editorial process.

- Response to the reviewers will be removed in this submission

8. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

- Cleaned document is submitted